Author's response to reviews

Title: Living Well with Diabetes: a randomized controlled trial of a telephone-delivered intervention for maintenance of weight loss, physical activity and glycaemic control in adults with type 2 diabetes

Authors:

Elizabeth G Eakin (e.eakin@sph.uq.edu.au)
Marina M Reeves (m.reeves@sph.uq.edu.au)
Alison L Marshall (alison.marshall@qut.edu.au)
David W Dunstan (David.Dunstan@bakeridi.edu.au)
Nicholas Graves (n.graves@qut.edu.au)
Genevieve N Healy (g.healy@sph.uq.edu.au)
Jonathan Bleier (j.bleier@uq.edu.au)
Adrian G Barnett (a.barnett@qut.edu.au)
Trisha O&M#8217;Moore-Sullivan (tmooore@cdcr.soms.uq.edu.au)
Anthony Russell (arussell@cdcr.soms.uq.edu.au)
Ken Wilkie (kcwilkie@ozemail.com.au)

Version: 2 Date: 30 July 2010

Author's response to reviews:

Formatting changes made as requested:
- Bullet points removed from tables
- All city and country details included in author affiliations
- 'Methods' renamed to 'Methods/Design' in both Abstract and Methods sections
- All figure and table placement text removed
- Underlining changed to bold
- All authors names added to references if under 30 authors
- Figure cropped to minimise white space
- Vertical lines removed from tables